\documentstyle[12pt]{article}

\textwidth6.25in
\textheight8.5in
\oddsidemargin.25in
\topmargin0in

\pagestyle{empty}

\begin{document}

\begin{center}
{\bf Dynamical Systems} \\
                        MA 760-1C, Spring 2005
\end{center}

\noindent Class meets (days? times?) in Seminar Room (458 CH)
\vspace*{5mm}

\noindent{\bf Instructor:}    Dr. Nikolai Chernov \\
{\bf Office:      } 492 A Campbell Hall, ph. 934-2154 or 934-8613 \\
{\bf E-mail: }       chernov@math.uab.edu\vspace*{3mm}

Class notes are available from the instructor's web page in PDF
format:
$$
{\rm http://www.math.uab.edu/chernov/teaching/teach.html} $$

\noindent{\bf Homework}: There are exercises in the classnotes,
which you need to do as homework. Ideally, you need to do (or at
least, figure out how to do) all the exercises. Write up your
solutions and submit them whenever possible. Overall, you need to
submit about 80\% of the exercises for full credit.\\

\noindent{\bf Presentations:} Periodically, students will present
some projects (usually, proofs of major theorems) in class. Every
student will need to present 2-3 projects during the semester.
\\

\end{document}
\end
